<mark>\_</mark>



# **Geophysical Research Letters**<sup>•</sup>

## **RESEARCH LETTER**

10.1029/2024GL109634

#### **Key Points:**

- City scale deployment of cool roofs leads to the greatest reduction in 2 m air temperature
- Green roofs do not decrease daily average temperature but have a daytime cooling effect
- Solar photovoltaic panels can reduce temperatures in London by capturing sensible heat flux and generate electrical power

#### **Supporting Information:**

Supporting Information may be found in the online version of this article.

#### Correspondence to:

O. Brousse, o.brousse@ucl.ac.uk

#### Citation:

Brousse, O., Simpson, C., Zonato, A., Martilli, A., Taylor, J., Davies, M., & Heaviside, C. (2024). Cool roofs could be most effective at reducing outdoor urban temperatures in London (United Kingdom) compared with other roof top and vegetation interventions: A mesoscale urban climate modeling study. *Geophysical Research Letters*, *51*, e2024GL109634. https://doi.org/10.1029/ 2024GL109634

Received 9 APR 2024 Accepted 7 JUN 2024

#### **Author Contributions:**

Conceptualization: O. Brousse, C. Simpson, C. Heaviside Data curation: O. Brousse Formal analysis: O. Brousse, C. Simpson Funding acquisition: M. Davies, C. Heaviside Investigation: O. Brousse, C. Simpson Methodology: O. Brousse, C. Simpson, A. Zonato, A. Martilli, C. Heaviside Project administration: C. Heaviside Software: O. Brousse, A. Zonato, A. Martilli Supervision: C. Heaviside Validation: O. Brousse Visualization: O. Brousse

© 2024. The Author(s).

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

## Cool Roofs Could Be Most Effective at Reducing Outdoor Urban Temperatures in London (United Kingdom) Compared With Other Roof Top and Vegetation Interventions: A Mesoscale Urban Climate Modeling Study

O. Brousse<sup>1</sup>, C. Simpson<sup>1</sup>, A. Zonato<sup>2</sup>, A. Martilli<sup>3</sup>, J. Taylor<sup>4</sup>, M. Davies<sup>1</sup>, and C. Heaviside<sup>1</sup>

<sup>1</sup>University College London, London, UK, <sup>2</sup>Koninklijk Nederlands Meteorologisch Instituut, Utrecht, The Netherlands, <sup>3</sup>Centro de Investigaciones Energéticas, Medioambientales y Tecnológicas, Madrid, Spain, <sup>4</sup>Tampere University, Tampere, Finland

**Abstract** Comprehensive studies comparing impacts of building and street levels interventions on air temperature at metropolitan scales are still lacking despite increased urban heat-related mortality and morbidity. We therefore model the impact of 9 interventions on air temperatures at 2 m during 2 hot days from the summer 2018 in the Greater London Authority area using the WRF BEP-BEM climate model. We find that on average cool roofs most effectively reduce temperatures ( $\sim$ -1.2°C), outperforming green roofs ( $\sim$ 0°C), solar panels ( $\sim$ -0.5°C) and street level vegetation ( $\sim$ -0.3°C). Application of air conditioning across London (United Kingdom) increases air temperatures by  $\sim$ +0.15°C. A practicable deployment of solar panels could cover its related energetic consumption. Current practicable deployments of green roofs and solar panels are ineffective at large scale reduction of temperatures. We provide a detailed decomposition of the surface energy balance to explain changes in air temperature and guide future decision-making.

**Plain Language Summary** Multiple common city scale passive and active interventions exist to reduce urban population's exposure to extreme heat during hot spells. Nonetheless, a proper comparison of the effect that each of these interventions may have on the temperatures experienced within large cities is missing. Additionally, the radiative and thermal mechanisms that lead to outdoor temperature changes are often not detailed and could lead to detrimental effects for local populations, such as indirect increase of water vapor or reflection of solar radiation. Our study, focusing over London, compares several common interventions through a modeling experiment and finds that cool roofs largely outperform other interventions during the two hottest days of the summer 2018. We also find that green roofs are ineffective on average and that solar panels and tree vegetation would only marginally change temperature exposures. Large scale deployment of air conditioning would lead to increased temperature in the core of London. Solar panels could potentially provide sufficient energy for running air conditioning all over London, creating comfortable indoor environments, and green roofs could reduce temperatures during the day. We argue that such inter-comparisons should guide future decision making.

## 1. Introduction

Cities have become a focus of attention for climate adaptation and mitigation solutions in the face of the increasing challenges induced by global urbanization and climate warming. More specifically, interest in urban temperatures has been growing, as cities are known to alter the local climate to which urban citizens are exposed (e.g., through the Urban Heat Island (UHI) effect (Oke, 1982; Oke et al., 2017) and thereby impact their thermal comfort and climate-related mortality (Arbuthnott et al., 2016; Gasparrini et al., 2015; Masselot et al., 2023). During hot spells, higher mortality and increased likelihood of thermal discomfort are generally observed, hence calling for adaptative and mitigative measures to reduce outdoor air temperatures at the city scale (Heaviside et al., 2016; Iungman et al., 2023; Salmond et al., 2016).

To respond to these risks to public health, passive and active strategies have been proposed to cool down the urban indoor and outdoor environments, and to provide indoor sheltering for the most vulnerable. Commonly proposed urban passive cooling strategies, which should be considered as measures that directly lower the temperatures, include increased urban vegetation, roofs incorporating vegetation (known as green roofs), or the deployment of



Writing – original draft: O. Brousse, C. Simpson, A. Zonato, A. Martilli, J. Taylor, M. Davies, C. Heaviside Writing – review & editing: O. Brousse, C. Simpson, A. Zonato, A. Martilli, J. Taylor, M. Davies, C. Heaviside highly reflective roofs that can be composed of different roofing materials (e.g., concrete, metal, or single-ply membrane), known as cool roofs (Broadbent et al., 2022; Santamouris et al., 2017). Changes to roofs can reduce indoor temperatures or cooling needs in a building (Virk et al., 2014). When deployed at city-scale, they can also reduce the outdoor air temperature and associated heat related mortality (H. Macintyre & Heavi-side, 2019). Concerning active strategies, meaning measures that transform the incoming radiation, into energy for example, or that mechanically change the temperatures, air conditioning (AC) is one of the most effective for protection against heat, although it can require large amounts of electrical power and increase outdoor temperatures (Salamanca et al., 2014; Stone Jr et al., 2021). Where passive strategies are not viable or insufficient, AC may be the only option, but it remains the primary solution in some high-income countries despite its problems, like in Phoenix, Arizona (Stone Jr et al., 2021). Lastly, while rooftop solar photovoltaic (PV) panels are primarily considered as a source of electrical power, they can also be considered as a passive-active strategy for impacting outdoor and indoor air temperature by increasing the roofs' albedo and by transforming incoming solar radiation into electrical power that can be used to run the AC system (Ma et al., 2017; Salamanca et al., 2016). Debate on their effectiveness as cooling strategies is currently ongoing (Sailor et al., 2021) and calls for more local studies to understand their costs and benefits at different latitudes.

In general, the impact of these strategies has been quantified for a variety of cities across the world (e.g., Chicago (Sharma et al., 2016), Madrid (Salamanca et al., 2014) or Singapore (Yang & Bou-Zeid, 2019)) and physical mechanisms that lead to the cooling/heating of the cities are well understood (Nazarian et al., 2022). Nonetheless, city-specific quantification of these impacts should still be performed to inform the general audience. In addition, few studies have explored in detail how these interventions impact the surface energy balance (SEB) and how these alterations at the city scale can relate to local changes in air temperatures (Broadbent et al., 2020; Krayenhoff et al., 2021). Furthermore, impact studies of urban scale interventions are often challenged by the lack of geographical data on realistic extent at which each intervention could be applied. This means that it is often impossible to give a proper estimate of the potential impact that interventions can have at a city-scale.

Large-scale controlled trials of urban interventions are, so far, rare, and much of the evidence for their effectiveness comes from modeling. Urban climate models (UCMs) embedded in regional climate models (RCMs), after proper model evaluation, can provide an accurate estimation of the impact of urban areas on local climates, as well as the impact of interventions (Krayenhoff et al., 2021). Indeed, most UCMs are coupled to a variety of intervention strategies (e.g., green roofs and solar panels (Zonato et al., 2021) or AC systems (Salamanca et al., 2010; Salamanca & Martilli, 2010)).

In this study, we use detailed urban climate modeling, as well as recent data on the actual coverage of passive (-active) interventions, to estimate the impact on outdoor temperature of several interventions during hot summer days on the average temperature within the Greater London Authority (GLA) boundaries. By considering a set of scenarios that include both active and passive heat adaptation strategies, our study makes a comprehensive analysis of the impact of each strategy on the outdoor air temperature and the SEB of the Greater London area. Because we expect some of these strategies to not be applicable to all buildings, like green roofs or solar PV panels, we provide practicable scenarios to compare the hypothetical maximum with more feasible implementation. We also discuss the potential energy costs and benefits of air conditioning and PV, and the advantages and disadvantages of vegetation-based strategies.

## 2. Methods

We run a set of 11 urban climate simulations to physically represent the potential impact of urban temperature reduction interventions at a high spatio-temporal resolution of 1 km with hourly outputs on the two hottest days of summer 2018, the 26th and the 27th of July—considering the 25th as spin-up time. 2018 is chosen as it was the hottest summer on average for the United Kingdom (McCarthy et al., 2019); see Appendix B in Supporting Information S1 for more information on the background weather conditions. We evaluated our control scenario—consisting of the same model set up as in Brousse et al. (2023) without any heat intervention strategy—over two hot summer days against a set of in situ observations of air temperature, relative humidity and wind speed and found its performance acceptable for our study (more information on the evaluation and its outcomes can be found in Appendix C in Supporting Information S1).

#### 2.1. Model Configuration

We use the Weather Research Forecast (WRF) v4.3 RCM following the previous model domain and set up of Brousse et al. (2023). The impact of cities on the local climate is simulated by activating the Building Energy Parameterization (BEP; Martilli et al. (2002)) model with its coupled Building Energy Model (BEM; Salamanca et al. (2010); Salamanca and Martilli (2010)), hereafter referred as BEP-BEM. We use the same model dynamics and physics set up as Brousse et al. (2023). For the urban canopy parameters and the land cover classification we use the default MODIS land cover from WRF and categorize the urban pixels to Local Climate Zones (LCZ; Stewart and Oke (2012)) following the WUDAPT-TO-WRF strategy to provide the WRF model with spatially explicit urban morphological parameters (Brousse et al., 2016; Demuzere et al., 2022). The building characteristics are provided per LCZ (see Figure D1A in Supporting Information S1) and are the same as in Brousse et al. (2023). More information on the model configuration can be found in Appendix A in Supporting Information S1.

Although our model is run over the whole of south-east England to accurately represent the interaction between local interventions and mesoscale atmospheric circulations, we focus our study on the GLA metropolitan area only because our practicable interventions are valid for that specific area. Our study aims at providing a general understanding of how different interventions can impact the average temperature of the GLA area during hot summer days. It does not investigate reasons explaining local impacts despite presenting some observable local features.

#### 2.2. Temperature Interventions

We define a set of 10 passive or active interventions that are theoretically applicable over the Greater London metropolitan area. Our scenarios are separated into two groups: (a) where interventions are deployed over the whole city to shed light on the maximum impact achievable by each intervention, and (b) practicable interventions that inform us about the actual potential for heat reduction in London—given the existing built environment. Most of our interventions consist of roof interventions and each intervention is compared to our control run to estimate the impact of each intervention on that urban climate. We also simulate the effect of widespread use of active cooling in the form of air conditioning.

Our scenarios consist of: (a) the deployment of cool roofs by increasing the albedo of each urban class to 0.85  $(CR_{100})$ ; (b) the deployment of green roofs composed of sedum vegetation in all urban classes by activating the green roof model from Zonato et al. (2021) with its default irrigation hours from 23.00 UTC to 01.00 UTC (GR<sub>100</sub>); (c) the deployment of solar PV panels on all urban roofs by activating the solar PV model from (Zonato et al., 2021) in its default version ( $PV_{100}$ ); (d) the installation of air conditioning systems in all buildings and at all floors by activating the air conditioning model embedded in the BEM model in its default version-with a coefficient of performance of 3.5 (see Salamanca et al. (2010); Salamanca and Martilli (2010);  $AC_{100}$ ); (e) the change of the urban vegetation from cropland and grass land to urban deciduous trees as implemented in Noah-MP (Niu et al. (2011),  $VG_{100}$ ); (f) the potential for solar PV deployment over roofs of existing buildings in the Greater London taking into account roof slope and overshadowing derived from the London solar opportunity map (PV<sub>poi</sub>; Steadman et al. (2020); Figure D1C in Supporting Information S1); (g) the potential for retrofit of green roofs on existing buildings in Greater London obtained through a detailed geographical information system analysis (C. Simpson et al. (2022); GR<sub>pot</sub>; Figure D1B in Supporting Information S1); (h) the installation of cool roofs in compact and open mid-rise areas (Local Climate Zones 2 and (e) from the center of London ( $CR_{mid}$ ); (i) in open low rise areas (Local Climate Zone 6; CR<sub>low</sub>) from the residential suburbs of London; and (j) in large lowrises of London (Local Climate Zone 8; CR<sub>1/r</sub>). More information about each model can be found in Appendix A in Supporting Information S1.

Potential conjoint use of green and cool roofs with solar panels have been proposed to increase solar panel efficiency while reducing urban surface temperatures in particular (Fleck et al., 2022; Schindler et al., 2018), but it was not possible to analyze this in our model. Sedum was modeled for the green roofs due to its increasing presence on new buildings in London (C. Simpson et al., 2022), but other plants such as turf may have a different impact.

To integrate the spatially-explicit variation in solar PV panels and green roofs coverage from the scenarios  $PV_{pot}$  and  $GR_{pot}$ , we adapted the code from Zonato et al. (2021) and added a field to the WRF geographical input files.



## Impact of intervention on air temperature at 2 m (T2) in the Greater London Authority (GLA) area















Figure 1. Average 2 m air temperature of the control run (T2; top left panel) and the average difference of each intervention ( $\Delta$ T2). Gray contours are WRF urban pixels (classified as an urban LCZ).

These two scenarios also explain why our study focuses only on GLA area as these are bounded to its administrative boundaries and are not applicable to the whole south-east of England which composes our WRF domain. Lastly, we expect the impacts to be more prominent in London due to its higher UHI intensity (Bassett et al., 2021; Brousse et al., 2022).

## 3. Results

#### 3.1. Impact of Interventions on the Temperature

In general, we find that cool roofs are the more effective interventions to reduce the heat in the GLA boundaries with a 2-day average cooling of ~1.2°C going up to ~2.0°C in certain locations (Figures 1 and 2). All cool roofs interventions contribute to a reduction of *T*2 on average (Figure 2) with major impacts found in the south and the east of GLA if deployed at 100% (CR<sub>100</sub>; Figure 1). Horizontal advection of cooling depends on the location where cool roofs are implemented. For instance, if cool roofs are implemented in more central and mid-rise areas of London, more open and low-rise residential areas appear to benefit from this cooling by ~0.15°C on average.





Impact of intervention on air temperature at 2 m (T2) and surface energy balance components in the Greater London Authority (GLA) area

Figure 2. Time-averaged impact of each intervention on each surface energy balance component and the respective change in air temperature at 2 m for all urban pixels (small dots). Average impact is represented by the large dots.

Interestingly, implementing cool roofs on residential low-rise buildings only ( $CR_{low}$ ) does not cool the central parts of the cities despite providing a substantial outdoor *T*2 reduction by up to ~2°C in the south-east. Establishing cool roofs only on large lowrises ( $CR_{llr}$ ) only results in a marginal heat reduction on average over London but does lead to a heat reduction of up to ~1°C in large lowrises areas of east-London. Compared to cool roofs, we find more minor decreases in *T*2 of ~0.3°C on average related to the change in natural cover from a mixture of croplands and grasslands to deciduous trees ( $VG_{100}$ ). The installation of solar panels over the whole city ( $PV_{100}$ ) leads to a reduction of *T*2 by ~0.5°C. Stronger cooling effects are found in the east of London for these two scenarios as well as for  $CR_{100}$  and  $CR_{llr}$ . Lastly, despite a high spatial heterogeneity of the cooling provided by the remaining interventions (green roofs ( $GR_{100}$  and  $GR_{pol}$ ); practicable retrofit potential of solar panels ( $PV_{pol}$ )), none is found to be beneficial for *T*2 reduction on average for the whole metropolitan area (Figures 1 and 2). Using air conditioning for maintaining the indoor temperature of all London's building stock to 21°C ( $AC_{100}$ ) induces a heating of central London by up to ~1°C and of ~0.15°C on average for the whole city.

In addition we find that, independent of the weather conditions, cooling brought by interventions tends to be greater in places that are hotter in the control run; except for  $CR_{llr}$  which shows greater cooling in places that are already cooler on average, and  $AC_{100}$  that would increase the heat in hotter parts of London (Figure D2 in Supporting Information S1). The latter could be explained by a higher release of sensible heat in the hottest parts of London under the  $AC_{100}$  scenario and an increase of albedo in large lowrises of  $CR_{llr}$  which are already cooler than other parts of London due to their location next to the rivers and canals and their higher roof reflective capacity.

These results highlight the complex interactions between local cooling capacity and the total cooling load that can be brought to a city by each intervention.





Impact of intervention on local (1 km<sup>2</sup>) air temperature at 2 m (T2) in the Greater London Authority (GLA)

Figure 3. Cumulative distributions of change in air temperature at 2 m for different intervention scenarios. Distributions are of change relative to the reference scenario in average daily mean, daily minimum, and daily maximum 2 m air temperature across grid cells. Distributions are from 200 bin histograms, normalized to unity and based on modeled air temperature on 26th and the 27th of July 2018 in London. The horizontal dashed line indicates the median change. The vertical dashed line indicates zero change.

#### 3.2. Temperature and Surface Energy Balance Changes

By relating the change in *T*2 of all urban pixels in the GLA area to their relative SEB components (Figure 2) we find that the large decrease in *T*2 induced by the full scale deployment of cool roofs (CR<sub>100</sub>) is linked to an increase in albedo by ~0.3 and a reduction of incoming net solar shortwave radiation by ~50 W m<sup>-2</sup>. This is accompanied by a reduction of sensible heat fluxes of ~30 W m<sup>-2</sup>, on average. Notably, we also observe an average increase of net longwave radiation budget by ~10 W m<sup>-2</sup> when deploying cool roofs over the whole city (CR<sub>100</sub>) and in open residential areas (CR<sub>*low*</sub>), which could be caused by less negative net longwave radiation during the day due to reduced upward emission from the lower temperature surface. Higher emissions of sensible heat fluxes by ~5 W m<sup>-2</sup> are found in both the air conditioning (AC<sub>100</sub>) and the deciduous tree (VG<sub>100</sub>) scenarios. While the former is expected as AC systems remove the accumulated indoor heat outside mostly in the form of sensible heat fluxes by ~12 W m<sup>-2</sup>. We however observe a decrease in surface albedo by 0.05 and an increase of latent heat fluxes by ~12 W m<sup>-2</sup>. We however observe a decrease in surface albedo by 0.05 and an increase of incoming net solar shortwave radiation of ~5 W m<sup>-2</sup>, on average, which could explain this phenomenon.

Despite an increase of *T*<sup>2</sup> induced by green roofs of less than 0.1°C on average, hourly spatial averages of changes in *T*<sup>2</sup> show that if green roofs were implemented across the whole city (GR<sub>100</sub>), temperatures could be decreased by ~0.5 to ~0.8°C during hot hours of the day (11.00 UTC to 14.00 UTC; Figure 3 and Figures D3 and D4 in Supporting Information S1). Nonetheless, temperatures would also be increased at times of daily minimum *T*<sup>2</sup> by ~0.5°C (03.00 UTC to 06.00 UTC, Figure 3). This is consistent with findings by Zonato et al. (2021) and explained by an increased release of latent fluxes during the day but a higher storage of heat fluxes at night due to green roofs' thermal mass. The current potential for green roofs in London (GR<sub>pol</sub>) would not reduce the temperature by more than 0.1°C on average across the period. During the hotter hours of the day, the impact of cool roofs (CR<sub>100</sub>) can be 2.5 times, and the one from cool roofs over open-low rises (CR<sub>low</sub>) 2.0 times, greater than the one of green roofs. Both these scenarios (CR<sub>100</sub> and CR<sub>low</sub>) also contribute to a constant reduction of temperature of at least 0.5°C throughout the day. This is correlated to a high reduction of emitted sensible heat fluxes and of the positive intakes of ground fluxes throughout daytime hours (Figures D3 and D4 in Supporting Information S1). Notably, the temperature reduction across the whole city induced by cool roofs over central mid-rises from London ( $CR_{mid}$ ) reaches a peak at later hours, around 15.00 UTC, a few hours after the peak in reduction of sensible heat fluxes reached at noon. Higher temperature reduction by ~0.8°C is observed in the evening if vegetation is changed to deciduous trees ( $VG_{100}$ ). During the day, no tangible impact is found in this case, as half of the urban pixels observe an increase in temperature while the other half observes a decrease. This may be because higher levels of sensible heat fluxes are emitted; more latent heat fluxes are also emitted throughout the day. Both full scale and practicable solar panel implementation ( $PV_{100}$  and  $PV_{pot}$ ) behave similarly to cool roofs by decreasing the emitted sensible heat fluxes and reducing the intake of fluxes to the ground during daytime hours. The temperature reduction of up to ~1.0°C, for the full-scale deployment, however, only happens during the late afternoon and the evening. At time of daily maximum temperatures,  $PV_{100}$  reduces the temperature in similar ways to  $GR_{100}$  (Figure 3 and Figures D3 and D4 in Supporting Information S1).

## 4. Discussion

Using the WRF BEP-BEM mesoscale urban climate over the Greater London metropolitan area during the two hottest days of 2018, we performed one of the first comprehensive evaluations of the impact of building and street level interventions tailored at reducing outdoor and indoor heat on the outdoor temperature and the SEB in London.

We find that interventions such as cool roofs, green roofs, rooftop solar PV, and changing natural spaces to deciduous woodland can contribute to a cooling of outdoor temperatures at hyper-local or city scales (also see Figures D5–D10 in Supporting Information S1, while widespread use of air conditioning could lead to substantial increases in outdoor temperature in the center of London. We find that this cooling is on average stronger in East London, suggesting that complex dynamics in atmospheric circulation and heat advection could be at play (see Brousse et al. (2022)) and would require further studies to disentangle when certain boroughs of London observe greater cooling reductions. This outcome could also simply be related to the short simulation period over which we average the results and be related to specific daily conditions (see Figure D11–D24 in Supporting Information S1). On average, cool roofs would be the most effective way of reducing outdoor temperature at the city scale, in line with other modeling studies (e.g., Sharma et al., 2016; Tan et al., 2023; Yang & Bou-Zeid, 2019). At the maximum, this temperature reduction is between 3.2°C and 2.8°C at times when daily spatial average temperatures reached 33°C and 37°C, respectively (Figure D4 in Supporting Information S1), similar to the cooling found in Phoenix and New York (Sinsel et al., 2021). In comparison, if green roofs or rooftop solar PV panels were deployed over all buildings in the Greater London area, the temperature reduction would have never reached more than  $\sim 1.0^{\circ}$ C for both days; considering that overnight, green roofs would increase the average temperature due to their increased heat-storage capacity that increases heat release at night (Tan et al., 2023). Of note, our green roofs are modeled with sedum vegetation as they are the most common in London; other designs may lead to other effects as Zonato et al. (2021) found an increased reduction using grass-based green roofs. Conversely, we find that changing the modeled vegetation type of urban natural spaces from grasslands/croplands to deciduous trees would reduce outdoor temperatures only during the night and found mixed effects during the day, possibly due to an increase in the surface albedo which leads to an increased sensible heat emission during the day. Latent heat would also increase as expected from more evapotranspiration and would be equivalent to the increase in latent heat from green roofs deployed at full scale. Although cooling via evapotranspiration reduces air temperature, it increases water vapor in the atmosphere; this could be beneficial for avoiding water stress of vegetation but affect the heat stress from city's inhabitants and should be evaluated using a set of multiple heat stress metrics (C. H. Simpson et al., 2023).

City scale impacts of any intervention are reduced when applied to a smaller area or fewer buildings. For cool roofs, we find that although major impacts are found when applied over low-rises residential areas of London, already applying them over central mid-rises of the city could reduce the temperature in surrounding residential areas. Because of the relatively small area covered by large lowrises in London, applying cool roofs over them would only marginally contribute to a heat reduction of the whole city, contradictory to results in Birmingham where commercial and industrial areas cover an important part of the city center (H. Macintyre & Heaviside, 2019). They could also be considered less effective as large lowrise buildings already observe higher albedo than more residential areas in our model (Brousse et al., 2023). Cool roofs may act to reduce indoor temperatures in summer, but also potentially in winter, thus increasing winter heating demand. Therefore, it is important that

they are installed in conjunction with insulation in homes (Taylor et al., 2018). However, in winter, cool roofs have less of a cooling effect on outdoor temperatures than they do in summer (H. L. Macintyre et al., 2021). While we find that full application of rooftop solar PV leads to some cooling on average, practicable application, constrained by existing building characteristics, is not enough to have an appreciable effect on the temperature of the GLA. Similarly, full application of sedum green roofs can reduce daytime temperatures, but the potential for retrofit of existing buildings with sedum green roofs is not enough to produce a substantial cooling effect at the city scale.

Although air conditioning would increase the outdoor temperature, it could still protect vulnerable populations from overheating indoors. Through our modeling experiment, we find that if solar panels were implemented in a practicable manner on the rooftops of existing buildings, the energy production, peaking at 15.15 MW at time of higher solar incidence (11.00 UTC) and with an average production of 4.70 MW, would cover the energetic consumption induced by AC systems which never goes above 8.78 MW during the hottest hours (15.00 UTC to 17.00 UTC), averaging to 4.61 MW throughout the day. At night, energy production being null by solar panels, the necessary energy for AC systems would have ideally been stored in batteries. Under the assumption that deployment of solar panels is not constrained by existing buildings, electric power production could go to 8.79 MW on average, and outdoor average temperature could be reduced by  $\sim 0.3^{\circ}$ C, something negligible in the practicable scenario. This temperature reduction should be carefully interpreted as several studies have found that solar panels could heat up the surrounding environment (Sailor et al., 2021). Nonetheless, these local impacts may be dependent on many factors, including the building height and the building plan area fraction (Zonato et al., 2021), or the increase in rooftop albedo. In our case, we argue that the capture of solar radiation by the solar PVs is sufficiently important to lower the expected intake of radiation by the building and that most of it is transformed in the form of electric energy rather than in sensible heat over the roof. In the evening and at night, the low thermal capacity of solar PVs explains the higher cooling capacity of the urban environment. In any case, our results call for more investigation on this specific topic as other studies, like (Tan et al., 2023), found a small night-time increase in temperature over Chicago when implementing solar PV over buildings' roofs. Hence mechanisms explaining solar PV efficiency in our model need to be further explored.

## 5. Limitations and Future Work

Several limitations apply to this study. First, we used only one UCM, and another UCM may produce different results depending on its dynamics and physics (Krayenhoff et al., 2021). Second, we focused our study only over two hot summer days due to computational limitations. In general, performing such comprehensive analysis with multiple scenarios comes at high computational costs, which often prevents the generalization of the results to expected average seasonal impacts of these interventions or to other cities at similar latitudes. Third, despite the fact that the 26th and the 27th of July were extremely hot days during the summer 2018 in the United Kingdom, full clear-sky conditions are absent from our model simulations. Some of the results observed in this study concerning the SEB may therefore simply be related to the overpass of clouds—although we did not find a particular difference in cloud coverage between our control simulation and our scenarios-and would be difficult to demonstrate as urban scale SEB products do not exist at hourly time step. Fourth, as discussed above, our study only focuses on outdoor temperatures and does not consider the benefits or the detriments caused by these interventions on other aspects (e.g., biodiversity, indoor temperature, increase in relative humidity, wind circulation, etc.). Fifth, the estimated impacts presented in this study assume a single type of application per intervention (e.g., green roofs are only composed of sedum vegetation, solar panels are based on a single type of photovoltaic materials, cool roofs are simply concrete with high albedo, for example) and do not explore the benefits provided by more recent technological advancement in each of these categories. For instance, our study did not look at the local climate effect of heat pumps turned on cooling mode as this feature is not yet available in BEP-BEM. Sixth, although evident average spatial features are observable for some interventions (e.g., increased temperature in the center of London under  $AC_{100}$ ) others may not have been captured due to random local effects dependent on weather conditions during the short period of study. Seventh, our  $VG_{100}$  scenario does not account for the radiative and thermal effects of trees within the urban canyon (like in BEP-Tree (Krayenhoff et al., 2020), for example,) but simply changes the bulk physical properties of the natural tile handled by the Noah-MP model. In the model, the natural tile in each model grid cell is subsequently weighted averaged with the urban tile, depending on the urban fraction. Lastly, all our estimated impacts should be interpreted cautiously as the average temperature differences resulting from the interventions are smaller than the model MAE with respect to

observations (Krayenhoff et al., 2021) and are dependent on 2 days with different cloud coverage and weather conditions. Despite these limitations, we expect our results to be at the least indicative of which interventions are capable of producing the greatest cooling effect in London during hot summer days.

To address the limitations of our study and push the research on the benefits of mitigative and adaptive strategies, future research should: (a) estimate the impact of each of these strategies on longer time periods (e.g., monthly or seasonal); (b) perform multi-model simulations over the same area to better quantify the uncertainty around the impact of each intervention; (c) investigate the complex interactions between local and city-scale impacts of tailored interventions; (d) estimate the costs related to the deployment of each of these interventions to perform cost-benefit analyses; (e) foster the deployment of urban scale SEB monitoring systems to validate model simulations—similar to crowd-sourcing of air temperature; (f) encourage the installation of monitoring systems of weather variables and SEB over more recent or recently retrofitted buildings which often integrate new roof designs, for example.

## 6. Conclusions

Adaptation to urban heat is of increasing priority. Our modeling study suggests that widespread adoption of cool roofs could be efficient at reducing urban temperatures, and therefore adverse health impacts, during hot spells. Rooftop solar PV may also provide a small amount of cooling as an additional benefit to power production; green roofs appear less effective in reducing temperatures but have other environmental benefits. Furthermore, relying on AC will lead to increased outdoor temperatures, despite improved thermal comfort indoors. It is reasonable to think that there will be a mixture of adaptation measures implemented across the city of London, and their applicability and impact should be further investigated. Our research is a first step toward this and will be of interest to city planners and decision-makers within the built environment sector.

## **Data Availability Statement**

The simulations done in this research were performed using the WRF model v4.4 (https://github.com/wrf-model/ WRF.git). The related outputs presented in this research and codes used to plot them are available at Brousse et al. (2024). The data used in the Supporting Information S1 for the study period selection and for model evaluation is publicly available. For ERA5 reanalysis data Copernicus users can download it from Hersbach et al. (2023). The open data from the Met Office Integrated Data Archive System (MIDAS) can be found at Met Office (2021). Information on how to gather Netatmo personal weather stations data can be found following the GitHub repository https://github.com/oscarbrousse/UHA\_CWS\_ERL2022.

#### References

- Arbuthnott, K., Hajat, S., Heaviside, C., & Vardoulakis, S. (2016). Changes in population susceptibility to heat and cold over time: Assessing adaptation to climate change. *Environmental Health*, 15(1), 73–93. https://doi.org/10.1186/s12940-016-0102-7
- Bassett, R., Janes-Bassett, V., Phillipson, J., Young, P. J., & Blair, G. S. (2021). Climate driven trends in London's urban heat island intensity reconstructed over 70 years using a generalized additive model. *Urban Climate*, 40, 100990. https://doi.org/10.1016/j.uclim.2021.100990
  - Broadbent, A. M., Declet-Barreto, J., Krayenhoff, E. S., Harlan, S. L., & Georgescu, M. (2022). Targeted implementation of cool roofs for equitable urban adaptation to extreme heat. *Science of the Total Environment*, 811, 151326. https://doi.org/10.1016/j.scitotenv.2021.151326 Broadbent, A. M., Krayenhoff, E. S., & Georgescu, M. (2020). Efficacy of cool roofs at reducing pedestrian-level air temperature during projected
  - 21st century heatwaves in Atlanta, Detroit, and Phoenix (USA). Environmental Research Letters, 15(8), 084007. https://doi.org/10.1088/1748-9326/ab6a23
    Brousse, O., Martilli, A., Foley, M., Mills, G., & Bechtel, B. (2016). WUDAPT, an efficient land use producing data tool for mesoscale models?
  - Integration of urban LCZ in WRF over Madrid. Urban Climate, 17, 116–134. https://doi.org/10.1016/j.uclim.2016.04.001
  - Brousse, O., Simpson, C., Kenway, O., Martilli, A., Krayenhoff, E. S., Zonato, A., & Heaviside, C. (2023). Spatially-explicit correction of simulated urban air temperatures using crowd-sourced data. *Journal of Applied Meteorology and Climatology*, 62(11), 1539–1572. https://doi. org/10.1175/jamc-d-22-0142.1
  - Brousse, O., Simpson, C., Walker, N., Fenner, D., Meier, F., Taylor, J., & Heaviside, C. (2022). Evidence of horizontal urban heat advection in London using six years of data from a citizen weather station network. *Environmental Research Letters*, 17(4), 044041. https://doi.org/10.1088/ 1748-9326/ac5c0f
  - Brousse, O., Simpson, C., & Zonato, A. (2024). Adaptation scenarios London 2018 (2-days) Dataset GRL. https://doi.org/10.5281/zenodo. 8333362
  - Demuzere, M., Argüeso, D., Zonato, A., & Kittner, J. (2022). W2W: A python package that injects WUDAPT'S local climate zone information in WRF. Journal of Open Source Software, 7(76), 4432. https://doi.org/10.21105/joss.04432
  - Fleck, R., Gill, R., Pettit, T., Torpy, F., & Irga, P. (2022). Bio-solar green roofs increase solar energy output: The sunny side of integrating sustainable technologies. *Building and Environment*, 226, 109703. https://doi.org/10.1016/j.buildenv.2022.109703
- Gasparrini, A., Guo, Y., Hashizume, M., Lavigne, E., Zanobetti, A., Schwartz, J., et al. (2015). Mortality risk attributable to high and low ambient temperature: A multicountry observational study. *The Lancet*, 386(9991), 369–375. https://doi.org/10.1016/s0140-6736(14)62114-0

## es Hajat, S o climate es-Basset d over 70

weather stations data.

Acknowledgments

CH is supported by a NERC fellowship

funding for the HEROIC project (216035/

Z/19/Z) from the Wellcome Trust, which

funds OB and CS. OB designed the study

and led the conception of the manuscript

with the support of CH and CS. OB was

responsible for the WRF modeling, the

model evaluation and the analysis. CS

provided support in the integration of

green roofs and solar PV inputs into WRF.

AZ implemented a spatially-explicit input

of solar PV and green roofs coverages in

WRF. AZ and AM offered guidance in the

set-up of the WRF model v4.4 and in the

provided support in the design of the study

and in the acquisition of knowledge about

manuscript. The authors thank Copernicus

MetOffice for the MIDAS Open data and

for the ERA5 reanalysis data, the UK

the Netatmo company for the personal

analysis of the outputs. JT and MD

building interventions. All authors

contributed to the writing of the

(NE/R01440X/1) and acknowledges

OA art

applicable Creat

- Heaviside, C., Vardoulakis, S., & Cai, X.-M. (2016). Attribution of mortality to the urban heat island during heatwaves in the west midlands, UK. Environmental Health, 15(S1), 49–59. https://doi.org/10.1186/s12940-016-0100-9
- Hersbach, H., Bell, B., Berrisford, P., Biavati, G., Horányi, A., Sabater, J. M., et al. (2023). ERA5 hourly data on single levels from 1940 to present. https://doi.org/10.24381/cds.adbb2d47
- Iungman, T., Cirach, M., Marando, F., Barboza, E. P., Khomenko, S., Masselot, P., et al. (2023). Cooling cities through urban green infrastructure: A health impact assessment of European cities. *The Lancet*, 401(10376), 577–589. https://doi.org/10.1016/s0140-6736(22)02585-5
- Krayenhoff, E. S., Broadbent, A. M., Zhao, L., Georgescu, M., Middel, A., Voogt, J. A., et al. (2021). Cooling hot cities: A systematic and critical review of the numerical modelling literature. *Environmental Research Letters*, 16(5), 053007. https://doi.org/10.1088/1748-9326/abdcf1
- Krayenhoff, E. S., Jiang, T., Christen, A., Martilli, A., Oke, T. R., Bailey, B. N., et al. (2020). A multi-layer urban canopy meteorological model with trees (BEP-Tree): Street tree impacts on pedestrian-level climate. Urban Climate, 32, 100590. https://doi.org/10.1016/j.uclim.2020. 100590
- Ma, S., Goldstein, M., Pitman, A., Haghdadi, N., & MacGill, I. (2017). Pricing the urban cooling benefits of solar panel deployment in Sydney, Australia. Scientific Reports, 7(1), 43938. https://doi.org/10.1038/srep43938
- Macintyre, H., & Heaviside, C. (2019). Potential benefits of cool roofs in reducing heat-related mortality during heatwaves in a European city. *Environment International*, 127, 430–441. https://doi.org/10.1016/j.envint.2019.02.065
- Macintyre, H. L., Heaviside, C., Cai, X., & Phalkey, R. (2021). Comparing temperature-related mortality impacts of cool roofs in winter and summer in a highly urbanized European region for present and future climate. *Environment International*, 154, 106606. https://doi.org/10.1016/ j.envint.2021.106606
- Martilli, A., Clappier, A., & Rotach, M. W. (2002). An urban surface exchange parameterisation for mesoscale models. Boundary-Layer Meteorology, 104(2), 261–304. https://doi.org/10.1023/a:1016099921195
- Masselot, P., Mistry, M., Vanoli, J., Schneider, R., Iungman, T., Garcia-Leon, D., et al. (2023). Excess mortality attributed to heat and cold: A health impact assessment study in 854 cities in Europe. *The Lancet Planetary Health*, 7(4), e271–e281. https://doi.org/10.1016/s2542-5196(23) 00023-2
- McCarthy, M., Christidis, N., Dunstone, N., Fereday, D., Kay, G., Klein-Tank, A., et al. (2019). Drivers of the UK summer heatwave of 2018. Weather, 74(11), 390–396. https://doi.org/10.1002/wea.3628
- Met Office. (2021). Midas open: UK hourly weather observation data, v202107. NERC EDS Centre for Environmental Data Analysis. https://doi.org/10.5285/3bd7221d4844435dad2fa030f26ab5fd
- Nazarian, N., Krayenhoff, E., Bechtel, B., Hondula, D., Paolini, R., Vanos, J., et al. (2022). Integrated assessment of urban overheating impacts on human life. *Earth's Future*, 10(8), e2022EF002682. https://doi.org/10.1029/2022ef002682
- Niu, G.-Y., Yang, Z.-L., Mitchell, K. E., Chen, F., Ek, M. B., Barlage, M., et al. (2011). The community Noah land surface model with multiparameterization options (Noah-MP): 1. Model description and evaluation with local-scale measurements. *Journal of Geophysical Research*, 116(D12), D12109. https://doi.org/10.1029/2010jd015139
- Oke, T. R. (1982). The energetic basis of the urban heat island. Quarterly Journal of the Royal Meteorological Society, 108(455), 1–24. https://doi.org/10.1002/qj.49710845502
- Oke, T. R., Mills, G., Christen, A., & Voogt, J. A. (2017). Urban climates. Cambridge University Press.
- Sailor, D., Anand, J., & King, R. (2021). Photovoltaics in the built environment: A critical review. *Energy and Buildings*, 253, 111479. https://doi.org/10.1016/j.enbuild.2021.111479
- Salamanca, F., Georgescu, M., Mahalov, A., Moustaoui, M., & Martilli, A. (2016). Citywide impacts of cool roof and rooftop solar photovoltaic deployment on near-surface air temperature and cooling energy demand. *Boundary-Layer Meteorology*, 161(1), 203–221. https://doi.org/10. 1007/s10546-016-0160-y
- Salamanca, F., Georgescu, M., Mahalov, A., Moustaoui, M., & Wang, M. (2014). Anthropogenic heating of the urban environment due to air conditioning. Journal of Geophysical Research: Atmospheres, 119(10), 5949–5965. https://doi.org/10.1002/2013jd021225
- Salamanca, F., Krpo, A., Martilli, A., & Clappier, A. (2010). A new building energy model coupled with an urban canopy parameterization for urban climate simulations—Part I. Formulation, verification, and sensitivity analysis of the model. *Theoretical and Applied Climatology*, 99(3– 4), 331–344. https://doi.org/10.1007/s00704-009-0142-9
- Salamanca, F., & Martilli, A. (2010). A new building energy model coupled with an urban canopy parameterization for urban climate simulations —Part II. Validation with one dimension off-line simulations. *Theoretical and Applied Climatology*, 99(3–4), 345–356. https://doi.org/10. 1007/s00704-009-0143-8
- Salmond, J. A., Tadaki, M., Vardoulakis, S., Arbuthnott, K., Coutts, A., Demuzere, M., et al. (2016). Health and climate related ecosystem services provided by street trees in the urban environment. *Environmental Health*, 15(1), 95–111. https://doi.org/10.1186/s12940-016-0103-6
- Santamouris, M., Ding, L., Fiorito, F., Oldfield, P., Osmond, P., Paolini, R., et al. (2017). Passive and active cooling for the outdoor built environment–analysis and assessment of the cooling potential of mitigation technologies using performance data from 220 large scale projects. *Solar Energy*, 154, 14–33. https://doi.org/10.1016/j.solener.2016.12.006
- Schindler, B. Y., Blaustein, L., Lotan, R., Shalom, H., Kadas, G. J., & Seifan, M. (2018). Green roof and photovoltaic panel integration: Effects on plant and arthropod diversity and electricity production. *Journal of Environmental Management*, 225, 288–299. https://doi.org/10.1016/j. jenvman.2018.08.017
- Sharma, A., Conry, P., Fernando, H., Hamlet, A. F., Hellmann, J., & Chen, F. (2016). Green and cool roofs to mitigate urban heat island effects in the Chicago metropolitan area: Evaluation with a regional climate model. *Environmental Research Letters*, 11(6), 064004. https://doi.org/10. 1088/1748-9326/11/6/064004
- Simpson, C., Brousse, O., & Heaviside, C. (2022). The potential of green roofs in London. In *Egu general assembly conference abstracts* (pp. EGU22–7671).
- Simpson, C. H., Brousse, O., Ebi, K. L., & Heaviside, C. (2023). Commonly used indices disagree about the effect of moisture on heat stress. NPJ Climate and Atmospheric Science, 6(1), 78. https://doi.org/10.1038/s41612-023-00408-0
- Sinsel, T., Simon, H., Broadbent, A. M., Bruse, M., & Heusinger, J. (2021). Modeling impacts of super cool roofs on air temperature at pedestrian level in mesoscale and microscale climate models. Urban Climate, 40, 101001. https://doi.org/10.1016/j.uclim.2021.101001
- Steadman, P., Evans, S., Liddiard, R., Godoy-Shimizu, D., Ruyssevelt, P., & Humphrey, D. (2020). Building stock energy modelling in the UK: The 3DStock method and the London building stock model. *Buildings and Cities*, 1(1), 100–119. https://doi.org/10.5334/bc.52
- Stewart, I. D., & Oke, T. R. (2012). Local climate zones for urban temperature studies. *Bulletin of the American Meteorological Society*, 93(12), 1879–1900. https://doi.org/10.1175/bams-d-11-00019.1
- Stone Jr, B., Mallen, E., Rajput, M., Broadbent, A., Krayenhoff, E. S., Augenbroe, G., & Georgescu, M. (2021). Climate change and infrastructure risk: Indoor heat exposure during a concurrent heat wave and blackout event in Phoenix, Arizona. Urban Climate, 36, 100787. https://doi.org/ 10.1016/j.uclim.2021.100787



- Tan, H., Kotamarthi, R., Wang, J., Qian, Y., & Chakraborty, T. (2023). Impact of different roofing mitigation strategies on near-surface temperature and energy consumption over the Chicago metropolitan area during a heatwave event. Science of The Total Environment, 860, 160508. https://doi.org/10.1016/j.scitotenv.2022.160508
- Taylor, J., Symonds, P., Wilkinson, P., Heaviside, C., Macintyre, H., Davies, M., et al. (2018). Estimating the influence of housing energy efficiency and overheating adaptations on heat-related mortality in the west midlands, UK. *Atmosphere*, 9(5), 190. https://doi.org/10.3390/ atmos9050190
- Virk, G., Jansz, A., Mavrogianni, A., Mylona, A., Stocker, J., & Davies, M. (2014). The effectiveness of retrofitted green and cool roofs at reducing overheating in a naturally ventilated office in London: Direct and indirect effects in current and future climates. *Indoor and Built Environment*, 23(3), 504–520. https://doi.org/10.1177/1420326x14527976
- Yang, J., & Bou-Zeid, E. (2019). Scale dependence of the benefits and efficiency of green and cool roofs. *Landscape and Urban Planning*, 185, 127–140. https://doi.org/10.1016/j.landurbplan.2019.02.004
- Zonato, A., Martilli, A., Gutierrez, E., Chen, F., He, C., Barlage, M., et al. (2021). Exploring the effects of rooftop mitigation strategies on urban temperatures and energy consumption. *Journal of Geophysical Research: Atmospheres*, 126(21), e2021JD035002. https://doi.org/10.1029/ 2021jd035002

## **References From the Supporting Information**

- Bougeault, P., & Lacarrere, P. (1989). Parameterization of orography-induced turbulence in a mesobeta-scale model. *Monthly Weather Review*, 117(8), 1872–1890. https://doi.org/10.1175/1520-0493(1989)117<1872:pooiti>2.0.co;2
- De Munck, C., Lemonsu, A., Bouzouidja, R., Masson, V., & Claverie, R. (2013). The GREENROOF module (v7. 3) for modelling green roof hydrological and energetic performances within TEB. *Geoscientific Model Development*, 6(6), 1941–1960. https://doi.org/10.5194/gmd-6-1941-2013
- Grassmann, T., Napoly, A., Meier, F., & Fenner, D. (2018). Quality control for crowdsourced data from CWS.
- Hammerberg, K., Brousse, O., Martilli, A., & Mahdavi, A. (2018). Implications of employing detailed urban canopy parameters for mesoscale climate modelling: A comparison between WUDAPT and GIS databases over Vienna, Austria. *International Journal of Climatology*, 38(S1), e1241–e1257. https://doi.org/10.1002/joc.5447
- He, C., Valayamkunnath, P., Barlage, M., Chen, F., Gochis, D., Cabell, R., et al. (2023). The community noah-mp land surface modeling system technical description version 5.0 (Tech. Rep.). NCAR Technical Note NCAR/TN-575+ STR. https://doi.org/10.5065/ew8g-yr95
- Jones, A., & Underwood, C. (2001). A thermal model for photovoltaic systems. Solar Energy, 70(4), 349-359. https://doi.org/10.1016/s0038-092x(00)00149-3
- Kendon, M., McCarthy, M., Jevrejeva, S., Matthews, A., Sparks, T., & Garforth, J. (2020). State of the UK climate 2019. International Journal of Climatology, 40(S1), 1–69. https://doi.org/10.1002/joc.6726
- Kendon, M., McCarthy, M., Jevrejeva, S., Matthews, A., Williams, J., Sparks, T., & West, F. (2023). State of the UK climate 2022. International Journal of Climatology, 43(S1), 1–83. https://doi.org/10.1002/joc.8167
- Napoly, A., Grassmann, T., Meier, F., & Fenner, D. (2018). Development and application of a statistically-based quality control for crowdsourced air temperature data. *Frontiers in Earth Science*, 6, 118. https://doi.org/10.3389/feart.2018.00118
- Rotach, M., Vogt, R., Bernhofer, C., Batchvarova, E., Christen, A., Clappier, A., et al. (2005). Bubble-an urban boundary layer meteorology project. *Theoretical and Applied Climatology*, 81(3–4), 231–261. https://doi.org/10.1007/s00704-004-0117-9
- Salamanca, F., Martilli, A., Tewari, M., & Chen, F. (2011). A study of the urban boundary layer using different urban parameterizations and highresolution urban canopy parameters with wrf. *Journal of Applied Meteorology and Climatology*, 50(5), 1107–1128. https://doi.org/10.1175/ 2010jamc2538.1
- Salamanca, F., Georgescu, M., Mahalov, A., Moustaoui, M., Wang, M., & Svoma, B. (2013). Assessing summertime urban air conditioning consumption in a semiarid environment. *Environmental Research Letters*, 8(3), 034022. https://doi.org/10.1088/1748-9326/8/3/034022
- Scherba, A., Sailor, D. J., Rosenstiel, T. N., & Wamser, C. C. (2011). Modeling impacts of roof reflectivity, integrated photovoltaic panels and green roof systems on sensible heat flux into the urban environment. *Building and Environment*, 46(12), 2542–2551. https://doi.org/10.1016/j. buildenv.2011.06.012
- Sunter, M. (2021). Midas data user guide for UK land observations, v20210705.
- Valencia, J. F. M., Henao, J., & Saher, R. (2023). The effect of removal of all non-functional turf in Las Vegas: Tradeoffs between water conservation, excessive heat, and storminess. (Tech. Rep.). *Copernicus Meetings*.
- Wouters, H., Demuzere, M., Blahak, U., Fortuniak, K., Maiheu, B., Camps, J., et al. (2016). The efficient urban canopy dependency parametrization (SURY) v1. 0 for atmospheric modelling: Description and application with the COSMO-CLM model for a Belgian summer. *Geo*scientific Model Development, 9(9), 3027–3054. https://doi.org/10.5194/gmd-9-3027-2016

19448007, 2024,